Author’s response to reviews

Title: Screening for Celiac Disease in Poorly Controlled Type 2 Diabetes Mellitus: Worth it or not?

Authors:

Muhammed Kizilgul (muhammedkzgl@gmail.com)
Ozgur Ozcelik (ozguroz_84@hotmail.com)
Selvihan Beysel (beyselfelvihan@gmail.com)
Hakan Akinci (hakanakinci289@hotmail.com)
Seyfullah Kan (seyfullahkan76@hotmail.com)
Bekir Ucan (uzm.dr.beikr@hotmail.com)
Mahmut Apaydin (drmahmut@yandex.com)
Mustafa Ozbek (mustafaozbek10@hotmail.com)
Erman Cakal (ermancakal@hotmail.com)

Version: 4 Date: 15 Jun 2017

Author’s response to reviews:

1. The FULL NAME of the ethics committee who approved your study in this section (Diskapi Teaching and Research Hospital (Date: 26.01.2015, Decree No:19/36) ) was added to material and method part

2. We wanted to add Dr.Mustafa Ozbek as an author since he helped us in preparing revisions. However, we were informed as helping in preparing of revisions of the manuscript is not enough to be an author. Hence, we removed his name from author list with his permission.